Author's response to reviews

Title: Psychiatric disorders and clinical correlates of suicidal patients admitted to a psychiatric hospital in Tokyo

Authors:

Naoki Hayashi (nhayashi55@nifty.com)
Miyabi Igarashi (crocus50@be.to)
Atsushi Imai (atsushi.imai0518@icom.home.ne.jp)
Yuka Osawa (sawako_smile@hotmail.com)
Kaori Utsumi (kumommmomm@gmail.com)
Yoichi Ishikawa (ginyu@fine.ocn.ne.jp)
Taro Tokunaga (p-tisiah@cool.odn.ne.jp)
Kayo Ishimoto (ggg10211360@yahoo.co.jp)
Hirohiko Harima (harima-tky@umin.ac.jp)
Yoshitaka Tatebayashi (tatebayashi-ys@igakuen.or.jp)
Naoki Kumagai (Naoki_Kumagai@member.metro.tokyo.jp)
Makoto Nozu (Makoto_Nozu@member.metro.tokyo.jp)
Hidetoki Ishii (z47783a@cc.nagoya-u.ac.jp)
Yuji Okazaki (okazaki@matsuzawa-hp.metro.tokyo.jp)

Version: 9 Date: 11 December 2010

Author's response to reviews:

Dear Sirs in the BMC Series Editorial Production Team

We present herewith the manuscript that we have incorporated some changes in at your request of the final check on 7 Dec 2010.

The changes we have made are as follows.

1. Authors’ affiliations and E-mail addresses have been updated.
2. Reference Nums 14 and 15 have been exchanged. (These references were arranged in a wrong order in the previous manuscripts.)
3. The last sentence in Background section has been shortened.

The previous sentence: “In the present study, we attempt to illuminate the diagnostic and clinical characteristics of this sample of suicidal patients, and to examine their associations with gender and age.”

The new one: “In the present study, we attempt to illuminate the clinical characteristics of this patient sample and their gender and age-relevance.” (The meaning has not been changed.)

We appreciate your beautiful edition work,
Sincerely,
Naoki Hayashi, MD, PhD.